Author's response to reviews

Title: Posterior reversible encephalopathy syndrome in a child with cyclical vomiting and hypertension: A case presentation

Authors:

Zakareya Gamie (ugm1zg@doctors.org.uk)
Akheel Rizwan (akheel929@googlemail.com)
Frances G Balen (Frances.Balen@midyorks.nhs.uk)
Michael Clarke (Michael.Clarke@leedsth.nhs.uk)
Mohammed M Hassoon (Mohammed.hassoon@midyorks.nhs.uk)

Version: 4 Date: 2 December 2010

Author's response to reviews:

Dear Editor,

Thank you for your e-mail dated 01/12/2010. We have now made the changes suggested and have attached a version of the updated manuscript.

Major revisions (We require the author to make these changes)

Comment 1 - Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Response - We have now indicated that all authors read and approved the final manuscript in the authors contribution section.

Comment 2 - References: If now possible, please provide the full citation details for reference 6. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references

Response - We have now provided the full details of reference 6.

 Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Comment 1 - Title page: Please remove the word count from the title page.

Response - We have now removed the word count.

Comment 2 - Headings: Please change the heading 'Case report' to 'Case presentation'.

Response - We have now changed the heading to Case presentation.

Comment 3 - Figure legends: Please move the figure legends to after the reference list.
Response - We have moved the figure legends to after the reference list.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Response - We have now deleted the following from the first sentence of the case report section: thought to be due to cyclical vomiting syndrome. We have corrected the typographical errors noticed.

Thank you

Dr Zakareya Gamie